International Journal of Finance (IJFIN)

Vol. 38, No. 4, July-August, 2025 pp. 25-65.

ISSN: 1041-2743

DOI: https://doi.org/10.5281/zenodo.15879473

Website: www.ijfin.com



Conventional Determinants of Dividend Payout Policy, Does Cross-Listing Matter?

Samar ElBanna¹, Ahmed Sakr², Heba Abdel-Gawad^{3,*}

^{1,2,3}Department of Finance, College of Management and Technology, Arab Academy for Science and Technology and Maritime Transport, Alexandria, Egypt.

¹ ORCID: 0009-0003-8078-9354 ² ORCID: 0000-0003-1026-1100 ³ ORCID: 0000-0002-3744-2975

*Corresponding Author: Heba Abdel-Gawad

Abstract

Corporate payout policy has garnered attention in corporate finance literature. Despite substantial study, opinions on the main factors affecting firm payout policy vary. Researchers have focused on developed markets and one dividend policy feature. Thus, this study examines how cross-listing, ownership structure, and firm-level characteristics like investment opportunity and free cash flow affect the dividend payout ratio and dividend decision for Egyptian cross-listed firms on the Egyptian and London stock exchanges. Compare the findings to single-listed firms exclusively listed on the Egyptian stock exchange. Control variables include leverage, profitability, liquidity, and firm size in this study. The sample for this study includes all Egyptian cross-listed and trading single-listed firms from 2007 to 2020. The results show that profitability and firm size affect cross-listed firms' dividend payout ratio. For single-listed firms, investment opportunity, profitability, and firm size affect dividend payout ratio. For the payout decision model, cross-listing, ownership structure, profitability, investment opportunity, leverage, and firm size are significant determinants for cross-listed firms,

while profitability, liquidity, and firm size are significant for single-listed firms. The findings reveal that cross-listing significantly affect the dividend decision.

Keywords: Dividend policy, Dividend decision, Dividend payout ratio, Cross-listing, Egypt.

How to Cite: Samar ElBanna, Ahmed Sakr, Heba Abdel-Gawad. (2025). Conventional Determinants of Dividend Payout Policy, Does Cross-Listing Matter?. *International Journal of Finance (IJFIN)*, 38(4), 25–65.

DOI: https://doi.org/10.5281/zenodo.15879473

1. Introduction

Dividend policy remains a contentious corporate finance issue. Financial economists have modelled corporate payout policies for over 50 years. Thirty years ago, [1] stated that the dividend image resembles a puzzle with unfit parts. A lot of literature has been written about dividend policy. However, [2] now agree with [1] that the "dividend puzzle" is a controversial issue in finance and financial economics, both in terms of share value and policy. The topic remains unresolved after 40 years of research.

Corporate dividend policy changes over time and between nations, especially emerging and developed financial markets. [3] Showed that emerging market dividend policies differed from developed ones. Emerging countries paid dividends at two-thirds the rate of developed nations. [4] Recently observed comparable declines in dividend rates within emerging economies. Organisations in emerging capital markets often lack the resources to fund their investments. Thus, they may need to rely largely on retained earnings thus lowering payout ratios.

In recent decades, cross-border barriers have decreased, making international capital markets more accessible. Cross-listing is a popular global capital-raising tool. This may be due to global stock exchanges' ability to attract stock listings and the company's assessment of cross-listing benefits [5-8].

Cross-listing can help a business raise capital, expand its investor base, improve visibility, liquidity, decreased flotation costs, and improve corporate governance [5-8]. These advantages may boost growth and dividend payout. Cross-listing enables firms raise

debt or equity to boost profits and consequently payout dividends. Recent research [5-8] shows that cross-listing increases dividend distributions. Since tourism and foreign investment deteriorated after the 2011 revolution which caused economic instability, Hard currency, especially dollars, is becoming scarcer. Therefore, Egyptian firms have turned to the international stock market to buy dollars for domestic operations.

Thus, Egyptian banks have stopped lending dollars to domestic businesses, especially those that make most of their money in pounds. The Cairo Stock Exchange is also criticized for its low liquidity and daily stock-price limit. Thus, GDRs have proven an efficient way to overcome these hurdles [9].

This study complements previous research investigating dividend payout determinants in emerging markets. The Egyptian stock market, which has been classified as a Secondary Emerging market, may be reclassified to unclassified market status because participants and index users have reported persistent delays in repatriating capital from Egypt since March 2023 [10].

The study's sample period is rich since it includes several political and economic events that affected the Egyptian stock market. These events include the 2008 global financial crisis, the 2011 and 2013 Egyptian revolutions, the 2016 Egyptian pound flotation, and the 2019 and 2020 Coronavirus pandemic. This research was able to evaluate payout policy determinants within a fertile sample period with shifting trends.

Furthermore, this study compares cross-listed firms' dividend payout ratio and dividend decision drivers to domestic single-listed firms. The payout ratio model shows that only profitability and firm size affect dividend size in cross-listed Egyptian firms, but investment opportunity, profitability, and firm size affect payout ratio in non-cross-listed Egyptian firms.

Regarding the payout decision model, in cross-listed firms, all variables are significant except liquidity, however in non-cross-listed firms; profitability, liquidity, and firm size affect payout decision. Conclusion: cross-listing affects dividend decision.

Rest of the paper is organized as follows: Section 2 presents corporate payout policy literature, including theoretical foundation and prior research. Section 3 shows this study's materials and methods. Section 4 presents the study's results. Section 5 shows the study conclusion.

2. Literature review

2.1. Corporate payout policy theories

Dividend policy couldn't be adequately explained by a single theory or determinant [11]. Common theories explaining payout policy includes the following:

2.1.1. The Bird in Hand Theory

The theory was originally proposed by [12]. It explains that investors frequently exhibit a preference for cash in hands, dividends, retained earnings, over future capital gains due to the unpredictability of future cash flows and their desire to minimize risk.

2.1.2. Tax Preference Theory

The theory argues that investors exhibit a preference for retained earnings in comparison to cash dividends. Due to the tax advantage of capital gains, investors might prefer a lower dividend payout to a higher one [13]. In Egypt, taxes on dividends and capital gains were implemented only in 2014; consequently, this theory is not relevant to our study.

2.1.3. Signaling Theory

According to [14] findings, a firm that possesses a greater degree of asymmetric information must distribute higher dividends in order to convey an equivalent level of earnings as a firm with a lesser degree of asymmetric information.

2.1.4. Agency Theory

[15] First proposed that conflicts of interest impact corporate payout policies, implying that dividends operate as a mechanism that aligns the interests of investors and managers. Two dichotomous agency dividend models have been constructed by [16].

First: **The substitution Model** According to [16], in firms with weak corporate governance measures, dividends serve as a substitute to poor legal protection for shareholders from management misuse of cash funds.

Second: **The outcome Model** According to [16], dividends are an outcome of effective legal protection of shareholders where minority shareholders are empowered to compel firms to distribute profits, thereby preventing insiders from misappropriating the earnings of the company.

2.1.5. Pecking order Theory

According to this theory, firms are more likely to finance new investments or dividends with retained earnings or internal sources of financing rather than external sources when dealing with a semi-strong efficient market [17, 18].

2.1.6. Life cycle Theory

The lifecycle theory of dividends posits that corporate dividend policy dependents on the stage that represent the firms' life cycles. Mature firms experience a decline in investment opportunities and growth rates, while simultaneously experiencing an increase in free cash flow and profitability. At this stage corporations start to pay dividends to shareholders in order to distribute their profits [19, 20].

2.2. Determinants of payout policy

This section provides an overview of the variables that have been examined in determining payout policy. These variables have been supported by previous studies, the research hypotheses are developed using the corporate payout theories addressed in section 1 as well as prior research.

2.2.1. Cross-listing

The bonding theory states that firms cross-list their shares on international markets to issue capital under better legal conditions. Managers can protect minority shareholders and prevent control exploitation by adhering to strict legal and corporate governance regulations, notably in US financial markets [21-25].

[26] Present empirical evidence supporting the "outcome model" that dividends are paid due to an effective shareholder protection scheme. Many researches support this, including [6-8].

Thus, local firms that cross-list their shares in overseas "more prestigious" markets may have a better dividend policy since they are more likely to safeguard minority shareholders rights due to the markets' strong governance standards. Based on the literature and agency theory, we claim that dividend payout and cross-listing are positively related.

H1: There is a significant positive relationship between Cross-listing and dividend payout policy.

2.2.2. Ownership structure

The nature of ownership within the company, along with the distribution of shares of ownership, undoubtedly influences the performance of firms and contributes to the reduction of agency costs.

As previously discussed dividend policy may serve to mitigate the agency problem and, consequently, act as a substitute for the ownership structure in addressing this matter.

Numerous researches have been conducted on the impact of ownership structure on dividend payout policy.

The first variable pertaining to ownership structure utilized in this study is managerial ownership. According to agency theory, an increase in managerial ownership aligns with a heightened interest among managers in pursuing more profitable projects, which consequently leads to a reduction in dividend payouts. This reasoning is supported by the findings of [27-29].

The second variable pertaining to ownership structure examined in this research is institutional ownership. [30] assert that agency costs may be mitigated by institutional investors through their monitoring roles; thereby distribution of dividends is unnecessary for alleviating agency-related issues. This adverse relation is supported by the findings of [31].

The final variable pertaining to ownership structure is free float ownership, as articulated within the framework of agency theory. [15, 32, 33] claim that dividends provide an indirect benefit of control to individual shareholders in situations where there is a lack of active monitoring of a firm's executives by its shareholders [33]. Consequently, the theory suggests a positive relationship between free float and payout policy.

In cross-listed firms, the enhanced corporate governance that results from cross-listing in foreign stock markets contributes to the increase in dividend policy in terms of likelihood, stability, and amount especially for firms that exhibit poor governance practices [8]. Consequently, firms with a higher concentration of ownership are more motivated to distribute higher dividends. This may be an indication of the desire of large shareholders like institutions to extract funds from the companies under their control [6].

It has been observed that cross-listed firms with higher individual (public) ownership pay lower dividends this is due to the fact that individual investors may prioritize capital gains over dividends and are more interested in long-term investments and short-term trading than in immediate profits. Since these investors do not rely on dividends to protect their rights, since cross-listing provides rigorous corporate governance measures for protecting shareholder rights [6, 7, 34].

Based on the preceding discussion regarding agency theory and the previous research conducted, the subsequent hypotheses regarding ownership structure are proposed:

H2: There is a significant negative relationship between managerial ownership and dividend payout policy in cross-listed firms.

H3: There is a significant positive relationship between institutional ownership and dividend payout policy of cross-listed firms.

H4: There is a significant negative relationship between free float and dividend payout policy of cross-listed firms.

2.2.3. Investment Opportunity

The pecking order theory argues that whenever a firm has residual earnings after funding all investment opportunities, those earnings ought to be distributed to shareholders in the form of cash dividends. This occurs because historically firms have preferred to use their internal resources to finance new investments; otherwise, they would not be able to pay dividends [35]. Both [26, 36] argue that the negative relationship between dividends and investment opportunities will be more significant when shareholder protection (which can be strengthened by cross-listing) and corporate governance are improved.

Numerous researches support this finding such as [6, 8]. Findings demonstrated that dividend size is lower in firms with higher investment opportunities, which suggests that firms prefer to retain more earnings when investment opportunities are available.

Relying on the pecking order theory and the supporting research, the following hypothesis can be predicted:

H5: There is a significant negative relationship between firm's investment opportunity and dividend payout policy in cross-listed firms.

2.2.4. Free cash flow

Free cash flow hypothesis posits that firms can mitigate the agency problem resulting from a surplus of free cash flows by paying dividends to shareholders [32]. This strategy is applied

instead of allowing management control of these cash flows, since they may misuse them and create serious agency costs. Cross-listing can augment the value of a company by reinforcing the emphasis on shareholders [37].

This is due to the fact that cross-listing makes it more challenging for managers to utilize corporate resources for their personal gains. It is plausible that cross-listed firms exhibit a higher degree of responsiveness to shareholder uncertainty regarding the possibility of misuse of surplus free cash flow, relative to their non-cross-listed counterparts. This argument is supported by research of [6, 8] findings which assert that free cash flow positively affects dividend payouts. Based on the aforementioned literature, the following hypothesis is predicted:

H6: There is a significant positive relationship between free cash flow and dividend payout policy in cross-listed firms.

2.3. Control variables

2.3.1. Leverage

According to [33] firms that possess significant financial leverage exhibit low payout ratios as a means to reduce the transaction costs associated with obtaining external financing. Furthermore, in accordance with [32] free cash flow hypothesis, dividends and debts serve as substitutes for monitoring the free cash flow that managers have access to.

This argument is supported by the majority of prior research, including [38-41]. Results indicate that cross-listed firms with high financial leverage tend to have lower dividend payouts. These findings are supported by [6-8]. Therefore, we predict a significant negative relationship between and dividend payout policy in cross-listed firms.

2.3.2. Profitability

However, it is not always the case that firms lacking dividend payments are unprofitable. In the event that a corporation determines that its internal growth prospects surpass investment opportunities accessible to shareholders elsewhere, it is advisable for the corporation to retain its earnings and reinvest them into the business. The majority of prior research, including that of [42-45], discovered a positive correlation between payout policy and profitability.

Cross-listing is a tool for increasing firm value as well as shareholder wealth since it enhances firm's ability to finance profitable projects, especially when the host market is the United States [46]. Findings of prior research show that more profitable cross-listed firms have higher dividend payouts than their non-cross-listed counterparts [6-8]. According to the aforementioned literature we predict a significant positive relationship between profitability and dividend payout policy in cross-listed firms.

2.3.3. Liquidity

According to [47] liquidity is the primary factor that influences dividend policy. An organization's capacity for paying dividends is enhanced when its overall cash position and liquidity are more robust. Moreover, in accordance with the agency theory, companies that possess substantial liquidity opt to distribute dividends as a means for overcoming agency problems. Such positive relationships have been supported by previous research, including that of [48, 49] which examined this area in depth.

According to [50] research using univariate and multivariate tests that cross-listing on US exchanges increases the proportion of earnings distributed to shareholders by firms originating from nations with inadequate shareholder protection. The aforementioned evidence aligns with the outcome theory, which posits a relationship between enhanced transparency and shareholder protection and dividend payout ratios.

Numerous researches support this finding such as [6-8]. Based on the aforementioned discussion, we predict a significant positive relationship between liquidity and dividend payout policy in cross-listed firms.

2.3.4. Firm size

The life cycle theory argues that during the early stages of a company's existence, it is probable that the acquired profits will be reinvested in efforts for promoting further growth. On the contrary, it is observed that mature firms tend to allocate a portion of their profits to investors in the form of dividends [51]. The aforementioned explanation of this relationship is supported by the majority of prior research, including [45, 52, 53].

Regarding the cross-listing context, [6-8] findings showed that larger cross-listed firms payout higher dividends, which supports the life cycle theory as well. Based on the above discussion, the following we predict a significant positive relationship between firm size and dividend payout policy in cross-listed firms.

3. Materials and Methods

First, descriptive statistics is used which is comprised of mean, maximum, minimum, and standard deviation, in order to describe the variables utilized in this study. Next, the generalized least square random effect regression analyses for panel data are conducted for the two models in this study. The first model tests the determinants of dividend payout policy by utilizing "payout ratio" as a dependent variable and the second model tests the "dividend decision" as a dichotomous dependent variable. Prior to testing the regression, regression diagnostic measures consisting of testing for normality, linearity, autocorrelation, multicollinearity, as well as the Hausman test are performed. Statistical techniques are implemented by using the STATA 18.0 software.

3.1. Variables description

Table 1. shows the research variables and its measurements

Variables	Indicator	Measurements	Reference
Dependent variables			
Dividend Decision	Div_Dec	Dummy Variable = 1 if the firm paid dividend or 0 otherwise	[42, 53]
Dividend Payout	Div_Payout	Dividend Per share / Earing Per share	[42]
Independent variables Cross-listing	Cross list	Dummy Variable = 1 if the firm paid dividend or 0 otherwise	
Managerial Ownership	_	Total Shares held by top management / total capital shares Total shares held by institution(s) / total capital shares Total shares held by external investors / total capital shares	[6-8]
Institutional Ownership	Mang_Own	Market value per share / book values per share	[69]
1		Cash flow from operation – capital expenditures	
Free Float	Inst Own	Short term debt/ total assets	[69]
Investment Opportunity	_		
	Freefloat	EBIT/T.Assets	[44, 59]
Free cash flow	Invest_opp	ROE = net profit / Total Equity ROA= net profit/Total Assets	[42, 45]
Control variables Leverage		Current assets/current liabilities Natural log of total assets	
	FCF		[70]
Profitability			
Liquidity	Lev		[71, 72]
Firm Size	Prof		
	Liq		[8, 65, 73]
	•		
	Size		[74, 75]
			[53,76]

3.2. Research model

```
Payout decision = \alpha i + \beta 1 * Cross - listing it + \beta 2 *

Managerial ownership it + \beta 3 * Institutional ownership it + \beta 4 *

Free float it + \beta 5 * Investment opportunity it + \beta 6 * Free cash flow + \beta 7 *

Short term debt it + \beta 8 * ROE it + \beta 9 * Current ratio it + \beta 10 *

Firm size it + \epsilon (2)
```

3.3. Sampling and Data Collection

This study used only secondary data. The Bloomberg Database provides the majority of research data, with annual reports from the Egyptian Company for Information Dissemination (EGID) filling gaps. This method ensured a balanced panel data set and reduced survivorship bias. The sample includes 175 Egyptian publicly traded firms. 13 are cross-listed enterprises from various sectors. To be included in the sample, cross-listed enterprises must have been cross-listed for at least five years previous to 2020 and have continuous dividend payout data for at least five years afterward. Inactive Egyptian stock market firms influenced the selection of 175 firms. Due to their inconsistent data across six years, these firms were eliminated from the sample. The appendix lists show excluded firms as well as Egyptian GDR firms utilised in this study. Banks and insurance firms are excluded from the sample due to their adherence to different disclosure rules and corporate governance procedures [77].

4. Results

4.1. Descriptive Statistics

Table 3 shows the descriptive statistics of all variables employed in this research.

Table 3. shows the descriptive statistics of research variables

Variable	Obs	Mean	Std. Dev.	Min	Max
Div Payout	2056	.393	.486	029	1.701
Div_Dec	2056	.703	.456	0	1
Mang_Own	1834	.011	.015	0	.05
Inst_Own	2450	.114	.219	0	.8
Freefloat	2093	.402	.232	.07	.855
Invest_opp	1779	1.339	.688	.602	3.205
FCF	2,450	.392	1.091	-1.160	4.654
Lev	2103	.359	.211	.055	.78
EBIT	2098	.073	.088	087	.263
ROE	1888	0.112	0.117	-0.163	0.411
ROA	1919	0.052	0.062	-0.096	0.235
Liq	2095	2.145	1.799	.468	7.619
Size	2137	1.886	.679	.772	3.213

The table shows descriptive statistics for the conventional and dependent variables used in the study from 2007 to 2020. The sample analysis includes 2,450 observations. Free cash flow (FCF) is operations cash flow minus capital expenditures. The Market-to-Book (MB) ratio serves as an indicator of investment opportunities (Invest_opp), calculated by dividing the market value per share by the book value per share. Leverage (Lev) is the ratio of short-term debt to total assets. Liquidity (Liq) is the ratio of Current assets to current liabilities. Profitability is the ratio of Earnings before Interest and Taxes (EBIT) to total assets. Firm size (Size) is the logarithm of total assets. Institutional, managerial, and free float percentages are Inst_Own, Mang_Own, and Freefloat, respectively. For dependent variables, dividend payout ratio, Div_Payout, is total cash dividends divided by earnings per share. The dividend decision, Div_Dec, is 1 if a dividend is distributed and 0 otherwise. All variables were winsorized at 99th percentile. The symbols ***, ***, and * indicate 1%, 5%, and 10% statistical significance.

4.2. Comparative statistics of cross-listed versus non cross-listed firms

In this section of the analysis, the researcher attempts to determine if cross-listed firms differ significantly from non-cross-listed firms in terms of payout ratio and other variables. Table 4 below shows the t-test results of the difference in means between cross-listed and non-

cross-listed firms for each variable that are being investigated in this research. The analysis conducted includes three levels of significance (1%, 5%, and 10%).

Table 4. Shows the t-test results of the difference in means between cross-listed and non-cross-listed firms

Variables	Cross-listed		Not cro	oss-listed	Diff.	P > t	Sig.
	Obs	Mean	Obs	Mean			
Mang_Own	76	0.02	1862	0.01	0.01	0.04	**
Inst_Own	85	0.14	2248	0.08	0.06	0.00	***
Free float	85	0.40	1820	0.39	0.01	0.04	**
Invest_opp	78	1.17	1701	1.51	0.34	0.07	*
FCF	45	4.25	929	3.32	0.93	0.46	
Lev	86	0.30	1883	0.36	-0.06	0.01	***
EBIT	76	0.09	1813	0.07	0.02	0.01	***
ROE	82	0.13	1806	0.11	0.02	0.10	*
ROA	76	0.07	1843	0.05	0.02	0.03	**
Liq	74	2.15	1771	1.78	0.37	0.00	***
Firm size	87	2.00	1843	1.87	0.13	0.04	**
Div_Payout	98	0.28	2069	0.16	0.12	0.00	***

This table reports umber of observation

And mean for the cross-listed and non-cross-listed subsample.

The "Difference test" column reports the t-test differences of means

Between cross-listed and non-cross-listed groups.

***, **, and * indicate significance at the 1%, 5% and 10%

Levels respectively,

4.3. Regression Analysis

4.3.1 Random-effect generalized least square estimates of payout ratio

Diagnostics are done before regression analysis. Table 5 shows how the link-test evaluates linearity between independent variables and log odds. The results show no specification error. Table 6 shows that no problems related to multicollinearity. The Hausman test determines whether to use random or fixed effects model.

Hausman test is insignificant for the 3 models; hence random effects model is used. Random-effect generalised least square analysis results are in Table 7. Model 1 evaluates the impact of variables, such as cross-listing, managerial ownership, institutional ownership, free float, investment opportunity, free cash flow, leverage, profitability, liquidity, and firm size, on dividend payout ratio.

The same variables affect dividend payout ratio for non-cross-listed corporations in Model 2. Model 3 explores how the same variables affect dividend payout ratio across the whole sample.

Firms who record a net loss while distributing dividends or pay dividends beyond their realised earnings are eliminated from the sample, which accounts for almost 16% of the total sample. Instead of distributing unrealised earnings, financially healthy firms maintain payout ratios that appropriately reflect their real earnings [78].

The upcoming results are derived from the application of the HAC estimator, which stands for "Heteroskedasticity and Autocorrelation Consistent Estimator." This methodology has been employed to address potential issues of heteroskedasticity and autocorrelation that may arise within the model.

Table 5. Linktest for specification error.

Dividend payout	Coefficient	Std. error	t	P>t	[95% conf.	interval]
_hat	1.46	0.37	3.91	0.00	0.73	2.20
_hatsq	-0.96	0.75	-1.28	0.20	-2.42	0.51
_cons	-0.04	0.04	-1.05	0.30	-0.13	0.04

This table shows the Linktest that is used to check if the gls regression used to test the effect of the dividend payout ratio on the variables (cross-listing, managerial ownership, institutional ownership, free float, investment opportunity, free cash flow, leverage, profitability, liquidity and firm size) suffers from any specifications error.

Table 6: Checking multicollinearity

Variable	VIF	1/VIF
Cross-listing	1.02	0.98
Mang_own	1.02	0.98
Inst_own	1.09	0.92
Free float	1.07	0.94
Invet_opp	1.03	0.97
FCF	1.19	0.84
Lev	1.51	0.66
ROE	1.16	0.86
Liq	1.45	0.69
Firms size	1.17	0.85

Tolerance and VIF and used to check if there is any linear combination between the independent variables of the model; cross-listing, Mang_own (managerial ownership),Inst_own (institutional ownership), free float, Invest_opp (investment opportunity), FCF (free cash flow), lev (leverage), ROE (profitability), Liq (liquidity) and firm size.

Table 7: shows the results of the panel data random effect of model (1)

	Model (1) Cross-listed			Model (2) Non-cross-listed		Model (3) Whole sample	
	Coef.	p-value	Coef.	p-value	Coef.	p-value	
Cross-listing		•		•	.136**	.048	
Mang Own	.265	.33	014	.88	.004	.967	
Inst Own	1.122	.848	.299	.749	.215	.815	
Free float	09	.821	.076	.22	.069	.254	
Invest_opp	095	.595	022**	.028	022**	.028	
FCF	.00027	0.521	0015**	.024	0016**	.022	
Lev	.155	.697	.054	.544	.056	.513	
ROE	1.268**	.025	.583***	.000	.593***	.000	
Liq	.014	.744	.011	.395	.01	.419	
Firm size	416**	.011	.061**	.017	.054**	.031	
Constant	1.034**	.015	077	.337	061	.428	
Obs.	18	3	572		590		
Wald Chi ²	16.6	58	48.	48.13		53.248	
Prob > chi ²	0.03	0.034		.000		.000	
Hausman test value	2.84	2.846		14.451		15.766	
Hausman p-value	.94	.944		.191		.172	
\mathbb{R}^2	.65	0	.142		.150		

The table displays the random effect model estimates of payout ratio regressed on cross-listing binary variable, management ownership ratio, institutional ownership ratio, free float ratio, investment opportunity, free cash flow, short-term debt, profitability ROE, liquidity and firm size as independent variables. Model (1) is cross-listed sample, Model (2) is non-cross-listed sample and Model (3) is the whole sample. The estimated coefficients and their p-values are shown separately for all firms, cross-listed firms, and non-cross-listed firms whereas ***, ** and * indicates statistical significance at the 1%, 5%, and 10% levels, respectively.

4.3.2 Random-effect generalized least square estimates of payout decision

After testing the effect of firm variables on dividend payout ratio using the random effect generalized least square regression. In this section, Models 1, Model 2 and Model 3 are tested again but this time using payout decision as a dependent variable.

Prior to executing the regression analysis, preliminary diagnostic assessments are conducted. The linearity of the independent variables and the log odds is assessed utilizing the link test, as demonstrated in Table 8.

Given that this study employs panel data, a Hausman test is performed to ascertain whether fixed effects or random effects models are more appropriate for analysis.

Based on the findings presented in Table 9, the Hausman test yields insignificant results for the three models under examination; therefore, it is advisable to employ the random effects model.

The subsequent results pertain to the application of the HAC estimator, or "Heteroskedasticity and Autocorrelation Consistent Estimator," which has been employed to resolve any heteroskedasticity or autocorrelation that might occur within the model. Table 9 presents the results of the Random-effect generalized least square analysis. Model 1 investigates the impact of variables, including cross-listing, managerial ownership, institutional ownership, free float, investment opportunity, free cash flow, leverage, profitability, liquidity and firm size on dividend decision for cross-listed firms. Model 2 investigates the impact of the same variables on dividend decision for non-cross-listed firms. Model 3 investigates the impact of the same variables on dividend decision for the whole sample.

Table 8: Linktest analysis for payout decision

Dividend decision	Coefficient	Std. error	t	P>t	[95% conf.	interval]
_hat	4.5	0.92	4.86	0.00	2.66	6.26
_hatsq	-2.35	0.62	-3.82	0.17	-3.56	-1.14
_cons	-1.22	0.34	-3.63	0	-1.88	-0.56

This table illustrates the Linktest that is utilised to check if the Gls regression employed to test the effect of the dividend payout decision on the variables (cross-listing, managerial ownership, institutional ownership, free float, investment opportunity, free cash flow, leverage, profitability, liquidity and firm size) suffers from any specifications error.

Table 9: shows the results of the panel data random effect of model (2)

		Model (4) Cross-listed		Model (5) Non-Cross-listed		Model (6) All Firms	
	Coef.	p-value	Coef.	p-value	Coef.	p-value	
Cross-listing					.196*	.075	
Mang_Own	305*	.075	.144	.347	.155	.269	
Inst_Own	17.576***	.000	549	.694	.052	.969	
Free float	603***	.007	135	.163	132	.164	
Invest_opp	221**	.041	001	.94	001	.942	
FCF	.000378	.940	.000748	.881	.000751	.883	
Lev	1.213***	.000	028	.85	.01	.947	
ROE	1.743***	.000	.564***	.000	.591***	.0000	
Liq	002	.951	.056***	.008	.051***	.01	
Firm size	367***	.000	.181***	.000	.169***	.000	
Constant	1.307***	.000	.253*	.069	.26**	.049	
Obs.	21		624		645		
Wald chi ²	92.06	5	45.07		48.84		
Prob > chi2	.000	.000		.000)	
Hausman test value	9.285	9.285		23.04		3	
Hausman p-value	.233	.233		.172		.195	
\mathbb{R}^2	.884		.128	3	.134	1	

Table shows payout decision random effect model estimates regressed on cross-listing binary variable, management ownership ratio, institutional ownership ratio, free float ratio, investment opportunity, free cash flow, short-term debt, profitability ROE, liquidity, and firm size as independent variables. Model (1) is cross-listed, Model (2) is non-cross-listed, and Model (3) is the complete sample. All firms, cross-listed firms, and non-cross-listed firms have separate estimated coefficients and p-values. ***, **, and * indicate statistical significance at 1%, 5%, and 10%, respectively.

5. Discussion

5.1. Descriptive Statistics

Table 3: presents the descriptive statistics for all variables employed in this study:

This examination will be conducted for the entire sample as well as for cross-listed and non-cross-listed firms in table 4. The objective is to investigate any significant difference in the means of the variables between the two categories of firms through t-test. Regarding dependent variables, the payout ratio exhibits an average of 40%, indicating that firms generally favor retaining a greater portion of earnings than disbursing dividends. This phenomenon may reflect the inclination of Egyptian firms to rely on internal financing within capital structure, attributed to the elevated interest rates prevailing in Egypt during the sample period.

The variables related to ownership structure indicate that the predominant form of ownership is free float, succeeded by institutional ownership and lastly managerial ownership. Upon examining the entire sample, firms in emerging markets present substantial investment opportunities; this is attributed to consistent growth and profitability [79]. Publicly listed Egyptian firms exhibit, on average, a positive free cash flow. This surplus may serve as an internal source of finance for firms that possess high investment opportunities. Egyptian firms exhibit a relatively low level of leverage, with an average of 36%.

The mean profitability of Egyptian firms is 7%, with a maximum recorded value of 26% and a minimum recorded value of -8.7%. This variation in profitability indicates diverse range of events that influenced firms throughout the sample period. Regarding liquidity, the mean value is 2, which falls within the ideal rule of thumb which posits that current ratio should be in the range between 1 and 2 [80].

Furthermore, the analysis of firm size indicates that the sample exhibits a mean of approximately 2, with a maximum value of 3. Finally, the skewness and kurtosis values of all independent variables indicate they do not follow a normal distribution. However, in accordance with the Gauss–Markov theorem, it is essential for obtaining "best linear unbiased estimators" (BLUE) that errors are uncorrelated, possess a mean of zero, and exhibit homoscedasticity. Thus, the normality assumption is not necessary to obtain BLUE results [81].

5.2 Comparative statistics of cross-listed versus not cross-listed firms.

Table 4 shows the comparison between cross-listed and non-cross-listed firms. The difference in ownership structure variables is significant. Investment opportunity of cross-listed firms is significantly lower since these firms are mature. Free cash flow does not differ significantly. The short-term debt ratio of cross-listed firms is 6% lower than that of non-cross-listed ones since cross-listed firms have less liquidity issues and do not rely heavily on short-term debt to finance its operations. In terms of profitability ratios ROA, ROE and EBIT, cross-listed firms outperform their non-cross-listed counterparts by 2% across all three indicators.

Moreover, cross-listed firms have a significantly higher liquidity; this may be attributed to the fact that cross-listing on international markets increases the firm's access to capital and its ability to raise funds, which reflects positively on the firm's financial performance. Cross-listed firms are slightly larger than non-cross listed counterparts. Finally, cross-listed firms payout more dividends than their non-cross-listed peers, as argued by [26] higher dividends is an outcome of high corporate governance standards and shareholder protection.

5.3 Random-effect generalized least square estimates of payout ratio

Table 7: Model (1) represents cross-listed firms, only profitability and business size have a significant relationship with dividend payout ratio.

Profitability of cross-listed Egyptian firms shows a significant positive relationship with dividend payout ratio. This result is consistent with the outcome model of the agency cost theory [26] which states that firms cross-list shares in overseas markets to benefits from stringent regulations and corporate governance measures as well as lower cost of external financing while reducing the risk of wealth expropriation leads to better valuation, higher profitability and consequently higher dividend payouts [26]. This result is supported by numerous previous studies including [6, 7, 8, 53, 82].

Results are related to Egyptian cross-listed firms since Cairo Stock Exchange is the subject of complaints due to its restricted daily limit on stock-price gains and losses, its lack of liquidity and the scarcity of foreign currency. Thus, firms cross-list shares overseas in highly liquid markets with cheap capital costs to finance profitable initiatives, which boosts firm profitability and dividend distributions.

Firm size shows a significant negative relationship with payout ratio. This result is consistent with the agency cost theory's substitution model, which states that greater oversight by financial intermediaries and improved legal protection through cross-listing protect shareholders and minimize agency costs. Large corporations with higher bonding pay fewer dividends [26]. This result is supported by several studies such as [7, 8, 26].

This result is related to Egyptian cross-listed firms, since these firms are large and profitable they seek global expansion. Thus, creating greater value to shares owned and financing growth opportunities are cross-listed firms' alternatives to dividend payouts.

In Model 2: which represents non-cross-listed firms, results revealed that Investment opportunity, free cash flow, profitability and firm size are the only variables that have a significant relationship with payout ratio.

Investment opportunity has a significant negative relationship with payout ratio. This is in line with the pecking order theory, which states that firms can only pay dividends when they have residuals after funding all of their investment opportunities [17]. This result is in line with various previous studies [42, 55, 56, 84].

This result may be attributed to the difficulty of external financing in the Egyptian market, which has experienced numerous crashes during the sample period. These crashes include the 2008 Global Financial Crisis, the 2011 Revolution, the 2016 currency devaluation, and the 2020 Coronavirus outbreak, which caused the Egyptian exchange to lose LE 134.1 billion of its market capitalisation [57]. Thus, the Egyptian stock market failures have made equity financing difficult.

Additionally, since the 2011 revolution, banks' lending rates have increased. It was 12% in 2011, 13% in 2016, and 18% in 2017 and 2018. It thereafter declined. Overall, external borrowing in Egypt is expensive, whether from banks or the stock market. This may explain why Egyptian firms prefer internal investment sources and perceive payouts as residual.

Free cash flow has a significant negative relationship with payout ratio, indicating that firms reduce dividends when free cash flow increases. This contradicts with the free cash flow theory, which states that firms pay dividends when they have extra free cash flow to reduce agency costs since it could be misused for management private benefits. This result is in line with some previous studies such as [58, 59, 84].

This justification fits the inefficient Egyptian market [61]. As mentioned in the preceding section, the Egyptian market restricts external financing. Second, when firms concern about future financial restrictions, management may hold onto cash reserves, which reduces dividend distributions [58].

Profitability has a significant positive relationship with payout ratio. This result is consistent with the life cycle theory, which posits that the more mature firms become, the less investment opportunities they possess, the more profitable they get and consequently the higher their dividend payouts [62]. This result is consistent with previous research of [42-44, 84].

This result is related to Egyptian firms which have less investment opportunities and are more mature, they are more profitable and they payout higher dividends. This is evident since 74% of dividend-paying firms in our sample generate positive net income. i.e.; they are profitable firms.

Firm size has a significant positive relationship with payout ratio. This is consistent with life cycle theory models [64, 85] that show firms distribute dividends when achieving optimal size. This result is consistent with numerous researches such as; [8, 65-67]. This result can be interpreted in Egypt since listed firms are large and have less growth opportunities due to maturity they payout higher dividends. This research shows that 74% of dividend-paying firms generate positive net income. i.e.; they are profitable firms.

In Model 3: which represents the whole sample, the significant variables affecting payout ratio are cross-listing, investment opportunity, Profitability and firm size. The first significant variable is cross-listing which shows a significant positive relationship with payout ratio. This result is consistent with the outcome model of the agency theory, according to which, dividends are the outcome of efficient corporate governance practices. Therefore, firms that exhibit reputational bonding pay higher dividends since they provide their shareholders with higher protection to pressure managers to distribute capital [26]. This result is supported by numerous researches such as; [6, 7, 26, 36, 50, 68].

This result can be interpreted in Egypt since Egyptian firms suffer from agency conflicts between shareholders and management due to poor corporate governance, firms adopted crosslisting to legally bond in prestigious markets which have strict corporate governance regulations and dividend payout rights. Thus, Egyptian cross-listed firms use dividends as an outcome of robust corporate governance system.

Regarding the significant variables; investment opportunity, profitability and firm size in model (3); the estimated outcomes are not considerably different from model (2), as the majority of observations pertain to non-cross-listed firms and very few to cross-listed ones. Thus, Model 2 and Model 3 regression analysis interpretations for these variables remain unchanged. The regression analysis found no variation in payout ratio variables between cross-listed and non-cross-listed firms. Therefore, cross-listing is irrelevant to the analysis.

5.4 Random-effect generalized least square estimates of payout decision

Table 9, Model (4) presents cross-listed firms, results show that managerial ownership, institutional ownership, free float, investment opportunity, leverage, profitability, and firm size affect payout decision.

Firstly, Managerial ownership has a significant negative relationship with dividend decision. This result is in line with agency cost theory, which claims that insider ownership can minimise agency costs, dividends are not used as a tool to protect shareholders' rights [30]. This result is supported by numerous researches [6-8, 69].

This result can be interpreted since cross-listing impose strict corporate governance regulations to protect shareholder rights and increase firm value besides higher managerial ownership could be an effective tool for governance as they are responsible for firm's financial decision, therefore, they will invest in profitable projects and hence no dividend decisions will be made.

The second significant variable is institutional ownership which shows a significant positive relationship with dividend decision. This result is consistent with the agency cost theory of [30], which states that institutional shareholders reduce agency conflicts through monitoring firm performance and management behavior. Thus, they use dividend payouts to reduce agency conflicts [87-89]. This outcome is consistent with previous research [90-93].

This result is interpreted since financial institutions are heavily regulated by the government this makes them cautious when investing in Egyptian cross-listed firms. Cross-listing's strict regulations can prevent financial institutions from self-serving and encourage them to prioritise minority shareholders' rights. Thus, the firm's dividend decision will be made i.e.; dividend payout will increase.

The third significant variable is free float. It shows a significant negative relationship with dividend decision. Free float result contradicts the assumptions of the agency theory. Dividends give owners indirect control in the absence of shareholder monitoring of insiders [15, 30, 33]. The theory predicts a positive relationship between free float and payout policy. However, in this study the free float in cross-listed firm is negatively related to payout decision. A 1% increase in cross-listed firm free float decreases dividend payout decision by 30%. This result is consistent with previous studies of: [6, 7, 34].

This result could be interpreted since in Egyptian cross-listed firms individual investors may prefer capital gains over dividends. These investors don't need dividends to protect their interests because cross-listing provides strong corporate governance.

The fourth significant variable is investment opportunity, which shows a significant negative relationship with dividend decision. This result is in line with the pecking order theory of [18] which argues that dividends are distributed to shareholders only when firms have residual earnings after financing all investment opportunities. This result is consistent with several researches such as: [6, 7, 42, 55, 56].

This result could be interpreted within the Egyptian cross-listed firms as follows: Banks in Egypt have been increasing their lending rates. It started 12% in 2011 reaching 18% 2018. It is evident that external financing in Egypt is expensive, whether from stock market or banks. Therefore, some Egyptian firms seek external finance through cross-listing. Therefore, cross-listing increases growth opportunities through public visibility, brand recognition, and lower capital costs [6]. Thus, firms allocate their resources in projects with the best growth opportunity rather than dividends.

The fifth variable that affects dividend decisions is leverage, which shows a significant positive relationship with payout decision. This result is consistent with the agency cost theory of [30] according to which, the agency problem between shareholders and creditors is a result of information asymmetry therefore; firms often expropriate creditors' wealth by increasing dividend payments. Numerous studies support this result for instance; [94-99].

This result could apply to Egyptian cross-listed firms since the relationship between debt issues and dividend payout is explained through reputation-building hypothesis, which argues that managers may not develop good reputation by dividend limitations, when firms have better access to credit due to tougher collateral laws and creditors have more credit information owing

to cross-listing. Thus, reputation building hypothesis suggests that legal rights and credit information improve dividend policy. I.e. more dividend decisions are made.

The sixth variable affecting dividend decision is profitability. Profitability has a significant positive relationship with dividend decision. It is consistent with the outcome model of the agency cost theory [26] that firms that cross-list their shares in foreign markets benefit from strict regulations, corporate governance policies, and lower external financing costs. This decreases wealth expropriation risk, increasing profitability, business valuation, and dividend distributions [26]. This finding is supported by many studies, including [6-8, 53, 82].

The same explanation given to the regression analysis of relationship between profitability and payout ratio will be applied here. The result affects Egyptian cross-listed firms since Cairo Stock Exchange is criticized for its strict daily stock-price gain and loss limit, limited foreign currency supply, and lack of liquidity. Thus, firms cross-list their shares in highly liquid international markets to finance profitable projects. This in turn, increases the firm's profitability and, as a result, managers make more payout decision.

Firm size has a significant negative relationship with payout decision. This outcome is in accordance with the substitution model of the agency cost theory, which posits that closer supervision by financial intermediaries and enhanced legal protection through cross-listing can be substituted for dividends. As a result, large corporations that have a higher level of bonding pay lower dividends [26]. Various researches support this finding such as [6-8].

The same explanation regarding regression analysis of payout ratio will be applied here. Since Egyptian cross-listed firms pursue international markets for better growth opportunities. Subsequently, cross-listed firms substitute the protection of shareholders' rights through dividend payments with the use of funds to finance growth opportunities and increase share value. Consequently, the more mature the firms get, the less they make a dividend decision.

In table 9, Model (5) which represents non-cross-listed firms, the results show that profitability, liquidity and firm size have a significant relationship with payout decision.

Profitability shows a significant positive relationship with dividend decision. This finding is in line with the life cycle theory, which argues that the more mature firms become, the fewer investment opportunities they possess, the more profitable they are and consequently, the higher their dividend payments [62]. This outcome is in line with the research conducted by [42-44, 84].

In the Egyptian context, this result is interpreted same as regression model of payout ratio: firms that have investment opportunities are likely to depend more on internal sources of finance, since external sources are costly and could hinder dividend payouts. Nevertheless, firms become more profitable and have fewer investment opportunities as they mature, thus making more dividend decisions. By examining the study sample, 74% of dividend paying firms have positive net income; meaning, they are profitable firms. This supports our findings.

Liquidity shows a significant positive relationship with dividend decision. This result is consistent with signaling theory which argues highly-liquid firms convey signals they are capable of paying obligations easily thus lower default risk [14, 102]. Consequently, managers make decisions to increase dividend payouts. This result is consistent with previous researches such as; [103, 104].

This result could be interpreted in the Egyptian context since it is evident that 74% of distributing firms have positive net income and most Egyptian listed firms are considered big sized firms, according to the life cycle theory, the more mature firms become, the fewer investment opportunities they have, the more profitable they are, the higher their liquidity, thus the higher their dividend decisions [67].

The last control variable is firm size which shows a significant positive relationship with dividend decision. This result is consistent with the life cycle theory of [63, 64] as they indicate firms distribute assets upon attaining an optimal size. This explains that large firms are more likely to distribute dividends than small ones. This outcome is in line with numerous studies, including [8, 65, 66].

This result can be interpreted in the Egyptian context same as in Model 2, table 4.3, since Egyptian listed firms are large and matured, they have less growth opportunities. Since Positive net income is shown in 74% of the dividend-paying firms in this study sample. Therefore, the more mature firms become, the more profitable they are, as a result, the more their dividend payout decisions [67].

Model (6); which includes all firms, cross-listing positively influences payout decision. Cross-listed firms are 20% more likely to make dividend decisions than non-cross-listed ones. These findings are consistent with the outcome model of the agency theory, which suggests that cross-listed firms are obligated to distribute dividends as an outcome of investors' rights protection [26]. This outcome is supported by numerous researches which includes; [6-8, 53,

82]. This result may be attributed to the difficulty of external financing in Egypt either from banks or stock market [57]. This motivated a number of firms to make cross-listing decision to overcome expensive financing, improve corporate governance and increase profitability which then encourage management to make dividend decisions.

With respect to significant variables such as profitability, liquidity, and firm size, the estimated results in model (6) are not significantly different from those in model (5). This is due to the fact that the majority of observations are related to non-cross-listed firms, while only a few ones are related to cross-listed firms.

Consequently, the regression analysis' interpretation for these variables in Models (5) and (6) remains unchanged. The regression analysis revealed a significant difference in the variables that influence payout decision between cross-listed and non-cross-listed firms. Therefore, cross-listing is relevant to the analysis.

6. Conclusions

This study analyses payout policies for cross-listed and single-listed firms in Egypt, a developing market with poor corporate governance and recurrent economic downturns. This article compares two dependent variables to independent and control variables: cross-listing, managerial ownership, institutional ownership, free float ownership, investment opportunity, leverage, profitability, liquidity, and company size. The payout ratio, which represents the amount of dividends that will be distributed from net income, is the primary dependent variable examined in this study. Dividend decision is the second dependent variable, determining whether to distribute dividends or not, regardless of quantity

According to the payout ratio model, two variables affect cross-listed firms' dividend size: profitability and firm size. We justify these results by observing that the Egyptian stock market's lack of liquidity and foreign currency supply raises financing costs. Cross-listing in highly liquid international markets offering hard currency supply and low capital costs which help firms fund profitable projects thus increasing firm's profitability and dividend payouts. As cross-listed firms mature, they replace shareholder rights protection through dividend payments with shareholder value created when they use funds in profitable ventures.

In single-listed Egyptian enterprises, investment opportunity, profitability, and firm size are significant. Since the study sample period has witnessed repeated market collapses and

rising Egyptian bank lending rates, stock market and bank loans are expensive. Thus Egyptian firms may prefer internal finance for investment opportunities and consider dividends residual. However, mature firms will distribute more dividends as they become more profitable and have fewer investment opportunities. This aligns with the life cycle theory that mature firms are more profitable and pay higher dividends.

The payout decision model in cross-listed firms shows that all variables for corporate payout decision are significant except liquidity. We justified this since effective corporate governance protects shareholder interests and cross-listing reduces manager-shareholder agency conflicts. As Managers' shares increase, their interests align with shareholders' to maximise profits and business value. Therefore, dividends are not used to safeguard shareholders' rights.

However, strict cross-listing regulations may prevent financial institutions from benefiting themselves and push them to protect minority shareholders. Consequently, firm's management will make a decision to payout dividends. Individual investors may value capital gains over dividends and long-term investments and trading over short-term earnings. Since cross-listing ensures strong corporate governance, these investors do not need dividends to protect shareholders' interests.

Under reputation building hypothesis, in cross-listing context, since firms have better access to credit due to stronger collateral rules and creditors have more credit information, therefore, corporate managers do not necessarily build good reputation through dividend restrictions. Consequently, Leverage positively affects dividend distribution. Egyptian firms used cross-listing to boost shareholder base, risk-sharing, and foreign currency access due to increased public visibility, brand awareness, and lower capital costs.

Thus cross-listing creates growth opportunities. Firms choose to invest in more profitable projects rather than pay dividends. However, when firms finance profitable projects and achieve higher profitability, managers are encouraged to pay dividends. As cross-listed firms mature, they replace dividend payouts "which has been a tool utilised to protect shareholder's rights in firms' profits" by maximising share value. Thus, managers do not make dividend decisions as firms mature.

Results demonstrate that profitability, liquidity, and firm size affect payout decisions for non-cross listed firms. Since Egyptian listed firms are large firms that have matured and experienced a drop in growth opportunities, they are more likely to have more liquidity because excess capital is not used to finance projects. Given that 74% of dividend-paying enterprises in this study sample have positive net income, or are profitable, encourages management to make dividend decisions.

Finally, we have discovered that cross-listing was relevant to the dividend payout decision but not to the dividend payout ratio model. This is due cross-listed firms have different dividend decision variables than non-cross-listed ones.

Abbreviations

GDRs: Global depository receipts

FTSE: Financial times stock exchange

EGID: Egypt for Information Dissemination

EGX: The Egyptian exchange

BLUE: Best linear unbiased estimators

HAC: Heteroskedasticity and Autocorrelation Consistent Estimator

EGX 30: Top 30 companies in terms of activity and liquidity in Egyptian exchange

EGX 100: Top 100 companies in terms of activity and liquidity in Egyptian exchange

COVID-19: Corona virus disease

Data Availability Statement

- 1. The data that support the findings of this study can be found at: https://www.egx.com.eg https://www.londonstockexchange.com
- 2. The data is available from the corresponding author upon reasonable request.

Conflicts of Interest

The authors declare no conflicts of interest.

Appendix

1. The names of the excluded firms from the sample are the following:

Abu Dhabi Islamic Bank - Egypt

Al Baraka Bank Egypt

Al Tawfeek Leasing Company-A.T.LEASE

Alexandria National Company for Financial Investment

Arab Moltaka Investments Co

Arabia Investments Holding

Aspire Capital Holding For Financial Investments

B Investments Holding

Banque Du Caire

Belton Financial Holding

CI Capital Holding For Financial Investments

Certificates of Odin Egyptian Equity Investment Fund-KASAB

Citadel Capital - Common Shares

Citadel Capital - Preferred Shares

Commercial International Bank (Egypt)

Contact Financial Holding

Credit Agricole Egypt

Delta Insurance

EDRs of Al Salam Holding Company

EGX 30 INDEX ETF

Egyptian Arabian (cmar) Securities Brokerage EAC

Egyptian Financial Group-Hermes Holding Company

Egyptian Gulf Bank

Egyptian Kuwaiti Holding

Egyptian Kuwaiti Holding-EGP

Egyptians Real Estate Fund Certificates

El Ahli Investment and Development

El Kahera El Watania Investment

El Orouba Securities Brokerage

Export Development Bank of Egypt (EDBE)

Faisal Islamic Bank of Egypt - In EGP

Faisal Islamic Bank of Egypt - In US Dollars

Gadwa For Industrial Development

Grand Investment Capital

Housing & Development Bank

International Company for Leasing (IncoLEASE)

Mohandes Insurance

Naeem Holding

National Bank Of Kuwait - Egypt - NBK

ODIN Investments

Orascom Financial Holding

Osool ESB Securities Brokerage

Pioneers Properties for Urban Development (PREDCO)

Prime Holding

Qatar National Bank Alahly

Raya Holding for Financial Investments

Saudi Egyptian Investment & Finance

Societe Arabe Internationale De Banque (SAIB)

Suez Canal Bank S.A.E

Fawry for Banking Technology and Electronic Payment

Integrated Diagnostics Holdings plc

Iron and Steel for Mines and Quarries

Rights Issue of El Obour Real Estate Investment-2

2. Egyptian GDR firms included in the sample are the following:

Egyptian Financial Group-Hermes Holding Company

Telecom Egypt

Ezz Steel

Lecico Egypt

Paint & Chemicals Industries (Pachin)

Edita Food Industries S.A.E

Orascom Investment Holding

Arabian Food Industries DOMTY

Madinet Nasr Housing

Alexandria Mineral Oils Company

GB AUTO

Commercial International Bank (Egypt)

Palm Hills Development Company

Amer group Holding

suez cement

Reference

- [1] Black, F. and Scholes, M. The effects of dividend yield and dividend policy on common stock prices and returns, Journal of Financial Economics. (1974), 1(1), 1-22. https://doi.org/10.1016/0304-405X(74)90006-3
- [2] Frankfurter, G., Kosedag, A., Schmidt, H. and Topalov, M. The perception of dividends by management, Journal of Psychology and Financial Markets. (2002), 3(4), 202-217. https://doi.org/10.1016/j.ribaf.2004.02.002
- [3] Glen, J., Karmokolias, Y., Miller, R. and Shah, S. "Dividend Policy and Behavior in Emerging Markets" IFC Discussion. (1995), Paper No. 26, www.ifc.org
- [4] Ramcharran, H. An Empirical Model of Dividend Policy in Emerging Equity Markets, Emerging Markets Quarterly 5. (2001), 39-49.
- [5] Wu, J. The effect of corporate governance and growth opportunities on dividend payout; does cross-listing matter? Master Thesis. Uppsala University. Sweden. 2018.
- [6] Cheng, Z., Cullinan, C. P., Liu, Z., & Zhang, J. Cross-listings and dividend size and stability: evidence from China, Accounting & Finance. (2021), 61(1), 415–465. https://doi.org/10.1111/acfi.12579
- [7] Mrzygłód, U., Nowak, S., Mosionek-Schweda, M., & Kwiatkowski, J. M. What drives the dividend decisions in BRICS countries?, Oeconomia Copernicana. (2021), 12(3), 593–629. https://doi.org/10.24136/OC.2021.020
- [8] Ghadhab, I. Bonding, signaling theory and dividend policy: Evidence from multinational firms, Journal of Asset Management. (2023), 24, 69–83. https://doi.org/10.1057/s41260-022-00289-7
- [9] American chamber of commerce in Egypt. The Stock Market. Available from:

- https://www.amcham.org.eg/publications/industry-insight/issue/49/the-stock-market. accessed (2001).
- [10] FTSE Russell. Classification of equity markets announcement. Available from: https://www.lseg.com/content/dam/ftse-russell/en_us/documents/country-classification-update-2024.pdf Accessed (2024).
- [11] Dewasiri, N.J. and Weerakoon, Y.K.B. Why do companies pay dividends? A comment, Journal of Corporate Ownership and Control. (2016), 13(2), 443-453. https://doi.org/10.22495/cocv13i2c2p5
- [12] Lintner, J. Distribution of Incomes of Corporations among Dividends, Retained Earnings, and Taxes, American Economic Review. (1956), 46(1), 97–113.
- [13] Brennan, M. Taxes, Market Valuation and Corporate Financial Policy, National Tax Journal. (1970), 23, 417-427. https://doi.org/10.1086/NTJ41792223
- [14] Bhattacharya, S. Imperfect Information, Dividend Policy, and "The Bird in the Hand" Fallacy. Bell Journal of Economics. (1979), 10, 259-270. https://doi.org/10.2307/3003330
- [15] Easterbrook, F. Two Agency-cost Explanations of Dividends, American Economic Review. (1984), 74, 650–59.
- [16] La Porta, Rafeal, Florenico Lopez-De-Silanes, Andrei Shleifer, and Robert W. Vishny. Agency Problems and Dividend Policies around the World, Journal of Finance. 2000a, 55, 1-33. https://doi.org/10.1111/0022-1082.00199
- [17] Myers, S. The Capital Structure Puzzle. Journal of Finance. (1984), 39, 574-592. https://doi.org/10.1111/j.1540-6261.1984.tb03646.x
- [18] Majluf, N., & Myers, S. Corporate Financing and Investment Decisions When Firms Have Information That Investors Do Not Have, Journal of Financial Economics. (1984), 13, 187-221. https://doi.org/10.1016/0304-405X(84)90023-0
- [19] Garengo, P., Nudurupati, S., & Bititci, U. Understanding the Relationship be- tween PMS and MIS in SMEs: An Organizational Life Cycle Perspective, Computers in Industry. (2007), 58, 677-686. https://doi.org/10.1016/j.compind.2007.05.006
- [20] O'Connor, T., & Byrne, J. Governance and the Corporate Life-Cycle, International Journal of Managerial Finance. (2015), 11, 23-43. https://doi.org/10.1108/IJMF-03-2013-0033

- [21] Coffee, Jack. The future as history: the prospects for global convergence in corporate governance and its implications, Northwestern University Law. (1999), 93, 641-708. https://dx.doi.org/10.2139/ssrn.142833
- [22] Doidge, Craig. US cross-listings and the private benefits of control: evidence from dual-class firms, Journal of Financial Economics. (2004), 72, 519-553. https://doi.org/10.1016/S0304-405X(03)00208-3
- [23] Doidge, Craig, G. Andrew Karolyi, René M. Stulz. Why do countries matter so much for corporate governance? Journal of Financial Economics. (2007), 86, 1-39. https://doi.org/10.1016/j.jfineco.2006.09.002
- [24] Reese, William A. Jr., and Michael S. Weisbach. Protection of minority shareholder interests, cross-listings in the United States, and subsequent equity offerings, Journal of Financial Economics. (2002), 66, 65-104. https://doi.org/10.1016/S0304-405X(02)00151-4
- [25] Stulz, René M. Globalization, corporate finance, and the cost of capital, Journal of Applied Corporate Finance. (1999), 12, 8-25. https://doi.org/10.1111/j.1745-6622.1999.tb00027.x
- [26] La Porta, R., F. Lopez-de-Silanes, A. Shleifer, R. Vishny. Agency Problems and Dividend Policies around the World, the Journal of Finance. (2000), 55(1), 1-33. https://doi.org/10.1111/0022-1082.00199
- [27] Al-Qahtani, T. H., & Ajina, A. The Impact of Ownership Structure on Dividend Policy the Evidence from Saudi Arabia. Journal of Emerging Issues in Economics, Finance and Banking. (2017), 6, 2187-2202.
- [28] Gul, F. A., Krishnamurti, C., Shams, S., & Chowdhury, H. Corporate Social Responsibility, Overconfident CEOs and Empire Building: Agency and Stakeholder Theoretic Perspectives. Journal of Business Research. (2020), 111, 52-68. https://doi.org/10.1016/j.jbusres.2020.01.035
- [29] Heba, A.-G., Ahmed, S., & Rabab, A. Conventional Determinants of Corporate Payout Policies in the Egyptian Stock Market. American Journal of Industrial and Business Management. (2021), 11, 1089-1112. https://doi.org/10.4236/ajibm.2021.1111066
- [30] Jensen, M. C., & Meckling, W. H. Theory of firm: Managerial behavior, agency costs and ownership structure, Journal of Financial Economics. (1976), 3(4), 305-360. https://doi.org/10.1016/0304-405X(76)90026-X

- [31] Al-Najjar, B. and Kilincarslan, E. The effect of ownership structure on dividend policy: evidence from Turkey, Corporate Governance. (2016), 16(1), 135-161. https://doi.org/10.1108/CG-09-2015-0129
- [32] Jensen, M. Agency Costs of Free Cash Flows, Corporate Finance, and Takeovers.

 American Economic Review. (1986), 76(1), 323–29.

 https://dx.doi.org/10.2139/ssrn.99580
- [33] Rozeff, M. S. Growth, Beta and Agency Costs as Determinants of Dividend payout Ratios, Journal of Financial Research. (1982), 5, 249-259. http://dx.doi.org/10.1111/j.1475-6803.1982.tb00299.x
- [34] Lam, K. C., Sami, H., & Zhou, H. The role of cross-listing, foreign ownership and state ownership in dividend policy in an emerging market, China Journal of Accounting Research. (2012), 5(3), 199-216. https://doi.org/10.1016/j.cjar.2012.06.001
- [35] Amidu, M. and Abor, J. Determinants of dividend payout ratios in Ghana, The Journal of Risk Finance. (2006), 7 (2), 136-145. https://doi.org/10.1108/15265940610648580
- [36] Mitton, T. Corporate governance and dividend policy in emerging markets, Emerging Markets Review. (2004), 5, 409–426. https://doi.org/10.1016/j.ememar.2004.05.003
- [37] Frésard, L., & Salva, C. The value of excess cash and corporate governance: Evidence from US cross-listings, Journal of Financial Economics. (2010), 98(2), 359-384. http://dx.doi.org/10.1016/j.jfineco.2010.04.004
- [38] Setiawan, D., Bandi, B., Phua, L. K., & Trinugroho, I. Ownership structure and dividend policy in Indonesia, Journal of Asia Business Studies. (2016), 10(3), 230-252. https://doi.org/10.1108/JABS-05-2015-0053
- [39] Ranajee, R., Pathak, R. and Saxena, A. To pay or not to pay: what matters the most for dividend payments? International Journal of Managerial Finance. (2018), 14(2), 230-244. https://doi.org/10.1108/IJMF-07-2017-0144
- [40] Wahjudi, E. Factors Affecting Dividend Policy in Manufacturing Companies in Indonesia Stock Exchange, Journal of Management Development. (2019), 39, 4-17. https://doi.org/10.1108/JMD-07-2018-0211
- [41] Basri, H. "Assessing determinants of dividend policy of the government-owned companies in Indonesia", International Journal of Law and Management. (2019), 61 (5), 530-541. https://doi.org/10.1108/IJLMA-09-2017-0215

- [42] Dewasiri, N., Banda, Y., Azeez, A., & Jayarathne, P. (2019). Determinants of Dividend Policy: Evidence from an Emerging and Developing Market. Managerial Finance, 45, 413-429. https://doi.org/10.1108/MF-09-2017-0331
- [43] Bokpin, G. Ownership Structure, Corporate Governance and Dividend Performance on the Ghana Stock Exchange, Journal of Applied Accounting Research. (2011), 12, 61-73. https://doi.org/10.1108/09675421111130612
- [44] Abdelsalam, O., & El-Masry, A. The Impact of Board Independence and Ownership Structure on the Timeliness of Corporate Internet Reporting of Irish-Listed Companies, Managerial Finance. (2008), 34, 907-918. https://doi.org/10.1108/03074350810915842
- [45] Al-Malkawi, H. Determinants of Corporate Dividend Policy in Jordan: An Application of the Tobit Model, Journal of Economic and Administrative Sciences. (2007), 23, 44-70. https://doi.org/10.1108/10264116200700007
- [46] Shmuel, B. G., Karolyi, A., & Lemmon, M. L. Multimarket Trading and Liquidity Theory and Evidence, The Journal of Finance. (2007), 62 (5), 2169 2200. https://doi.org/10.1111/j.1540-6261.2007.01272.x
- [47] Baker, H. K., Farrelly, G. E., & Edelman, R. B. A Survey of Management Views on Dividend Policy, Financial Management. (1985), 14, 78-84. https://doi.org/10.2307/3665062
- [48] Khan, N.U., Burton, B.M. and Power, D.M. Managerial views about dividend policy in Pakistan, Managerial Finance. (2011), 37(10), 953-970. https://doi.org/10.1108/03074351111161600
- [49] Kumar, B. R., & Waheed, K. A. Determinants of Dividend Policy: Evidence from GCC Market, Accounting and Finance Research. (2015), 4, 17-29. https://doi.org/10.5430/afr.v4n1p17
- [50] Petrasek, Lubomir, Do Transparent Firms Pay Out More Cash to Shareholders? Evidence from International Cross-listings, Financial Management. (2012), 41(3), 615-36. https://doi.org/10.1111/j.1755-053X.2012.01192.x
- [51] Denis, D., Osobov, I.Why Do Firms Pay Dividends? International Evidence on the Determinants of Dividend Policy, Journal of Financial Economics (2008), 89, 62-82. https://doi.org/10.1016/j.jfineco.2007.06.006

- [52] Yusof, Y., & Ismail, S. Determinants of Dividend Policy of Public Listed Companies in Malaysia. Review of International Business and Strategy. (2016), 26, 88-99. https://doi.org/10.1108/RIBS-02-2014-0030
- [53] Takmaz, S., Evrim Mandaci, P., & Durukan, M.B. The Impact of Catering Incentives on the Dividend Policy: Evidence from Turkish Firms, Managerial Finance. (2021), 47, 897-914. https://doi.org/10.1108/MF-04-2020-0215
- [54] Myers, S.C. and Majluf, N.S. Corporate financing and investment decisions when firms have information that investors do not have, Journal of Financial Economics. (1984), 13(2), 187–221. https://doi.org/10.1111/j.1540-6261.1984.tb03646.x
- [55] Imamah, N., Lin, T., Suhadak, Handayani, S., & Hung, J. Islamic Law, Corporate Governance, Growth Opportunities and Dividend Policy in Indonesia Stock Market, Pacific-Basin Finance Journal. (2019), 55, 110-126. https://doi.org/10.1016/j.pacfin.2019.03.008
- [56] Khana, Z., Ramakrishnanb, S., Malikc, M. F., & Ullahd, M. (2020). Impact of Growth Opportunities on Dividend Policy: Evidence on the Role of CEO Duality in the Emerging Economy of Pakistan. International Journal of Innovation, Creativity and Change, 11, 783-794.
- [57] OECD (2020). The Covid-19 Crisis in Egypt. Available from: https://www.oecd.org/mena/competitiveness/The-Covid-19-Crisis-in-Egypt.pdf
- [58] Almeida, H., Campello, M., & Weisbach, M. S. The cash flow sensitivity of cash, The Journal of Finance. (2004), 59(4), 1777-1804.https://doi.org/10.1111/j.1540-6261.2004.00679.x
- [59] Mirza, H. H., & Azfa, T. Ownership structure and cash flows as determinants of corporate dividend policy in Pakistan, International Business Research. (2010), 3(3), 210-221.https://doi.org/10.5539/ibr.v3n3p210
- [60] Deng, L., Li, S., Liao, M., & Wu, W. Dividends, Investment and Cash Flow Uncertainty: Evidence from China, International Review of Economics & Finance. (2013), 27, 112-124. https://doi.org/10.1016/j.iref.2012.09.005
- [61] Metwally, A. H., & Darwish, O. Evidence of the Over Confidence Bias in the Egyptian Stock Markets in Different Market States, the Business and Management Review. (2015), 6, 178-198.
 - [62] Mueller, D.C. A Life Cycle Theory of the Firm, the Journal of Industrial Economics. (1972), 20(3), 199-219. https://doi.org/10.2307/2098055

- [63] Stiglitz, J. (1973). Taxation, corporate financial policy, and the cost of capital, J. Public Econ. 2, 1–34. https://doi.org/10.1016/0047-2727(73)90008-X
- [64] Sinn, H.-W. The vanishing Harberger triangle, Journal of Public Economy. (1991), 45, 271–300. https://doi.org/10.1016/0047-2727(91)90029-2
- [65] Barros, V., P.V. Matos, and J.M. Sarmento. What firm's characteristics drive the dividend policy? A mixed-method study on the Euronext stock exchange, Journal of Business Research. (2020), 115, 365–377.
 https://doi.org/10.1016/j.jbusres.2019.11.042
- [66] Farinha, J. Dividend policy, corporate governance, and the managerial entrenchment hypothesis: An empirical analysis, Journal of Business Finance & Accounting. (2003), 30(9–10), 1173–1209. https://doi.org/10.1111/j.0306-686X.2003.05624.x
- [67] Ahmed, I. E. Factors Determining the Selection of Capital Budgeting Techniques, Journal of Finance and Investment Analysis. (2013), 2, 77-88.
- [68] Adjaoud, F., & Ben-Amar, W. "Corporate governance and dividend policy: shareholders' protection or expropriation?" Journal of business finance & accounting. (2010), 37(5-6), 648-667. https://doi.org/10.1111/j.1468-5957.2010.02192.x
- [69] Dhuhri, R., & Diantimala, Y. The Influence of Institutional Ownership, Individual Ownership, and Managerial Ownership toward Dividend Payout Ratio at Non-Financial Companies Registered in Indonesia Stock Exchange in 2012-2016, International Journal of Social Science and Economic. (2018), 3(3), 786-801.
- [70] Islam, S. M. T., Ghosh, R., & Khatun, A. Slack Resources, Free Cash Flow and Corporate Social Responsibility Expenditure: Evidence from an Emerging Economy. Journal of Accounting in Emerging Economies. (2021), 11, 533-551. https://doi.org/10.1108/JAEE-09-2020-0248
- [71] Setiawan, D., Bandi, B., Phua, L., & Trinugroho, I. Ownership Structure and Dividend Policy in Indonesia, Journal of Asia Business Studies. (2016), 10, 230-252. https://doi.org/10.1108/JABS-05-2015-0053
- [72] Ramadan, I. Z. The Jordanian Catering Theory of Dividends. International Journal of Business and Management. (2015), 10, 226-230. https://doi.org/10.5539/ijbm.v10n2p226
- [73] Trabelsi, D., Aziz, S., & Lilti, J.-J. A Behavioral Perspective on Corporate Dividend Policy: Evidence from France. Corporate Governance. (2019), 19, 102-119. https://doi.org/10.1108/CG-02-2018-0077

- [74] Al-Kayed, L.T. "Dividend payout policy of Islamic vs conventional banks: case of Saudi Arabia", International Journal of Islamic and Middle Eastern Finance and Management. (2017), 10(1), 117-128. https://doi.org/10.1108/IMEFM-09-2015-0102
- [75] Budagaga, A.R. "Determinants of banks' dividend payment decisions: evidence from MENA countries", International Journal of Islamic and Middle Eastern Finance and Management. (2020), 13 (5), 847-871. https://doi.org/10.1108/IMEFM-09-2019-0404
- [76] Sheikh, S. Corporate social responsibility and corporate payout policy: the impact of product market competition, International Journal of Managerial Finance. (2020). https://doi.org/10.1108/IJMF-09-2020-0465
- [77] Baker, M., & Wurgler, J. "A catering theory of dividends", The Journal of Finance. (2004), 59(3), 1125-1165. https://doi.org/10.1111/j.1540-6261.2004.00658.x
- [78] Harackiewicz, J., Barron, K., Pintrich, P. R., Elliot, A., & Thrash, T. "Revision of achievement goal theory: Necessary and illuminating", Journal of Educational Psychology. (2002), 94(3), 638–645.

 https://psycnet.apa.org/doi/10.1037/0022-0663.94.3.638
- [79] Walsh, J. P., Kress, J. C., & Beyerchen, K. W. "Promises and perils at the bottom of the pyramid", Administrative Science Quarterly. (2005), 50(3), 473-482. https://doi.org/10.2189/asqu.2005.50.3.473
- [80] Panigrahi, A. K. "Liquidity Management of Indian Cement Companies: A Comparative Study". IOSR Journal of Business and Management. (2013), 14(5), 49-61.
- [81] Schmidt, A. F., & Finan, C. (2018). Linear Regression and the Normality Assumption.

 Journal of Clinical Epidemiology, 98, 146-151.

 https://doi.org/10.1016/j.jclinepi.2017.12.006
- [82] Ghadhab, I., & M'rad, M. Does U.S. cross-listing come with incremental benefit for already U.K. cross-listed firms? The Quarterly Review of Economics and Finance. (2018), 69, 188–204. https://doi.org/10.1016/j.qref.2018.02.002
- [83] Sakir, A., & Fadli, M. Influence of Managerial Ownership, Debt Policy, Profitability, Firm Size, and Free Cash Flow on Dividend Policy: A Study in Indonesian Manufacturing Industries, Delhi Business Review. (2014), 15, 15-22. https://doi.org/10.51768/dbr.v15i1.151201402

- [84] Heba, A.-G., Ahmed, S., & Rabab, A. Conventional Determinants of Corporate Payout Policies in the Egyptian Stock Market. American Journal of Industrial and Business Management. (2021), 11, 1089-1112. https://doi.org/10.4236/ajibm.2021.1111066
- [85] Stiglitz, J. Taxation, corporate financial policy, and the cost of capital, J. Public Economics. (1973), 2, 1–34. https://doi.org/10.1016/0047-2727(73)90008-X
- [86] Deng, L., Li, S., Liao, M., & Wu, W. Dividends, Investment and Cash Flow Uncertainty: Evidence from China. International Review of Economics & Finance. (2013), 27,112-124. https://doi.org/10.1016/j.iref.2012.09.005
- [87] Demsetz, H. and Lehn, K. The Structure of Corporate Ownership: Causes and Consequences, Journal of Political Economy. (1985), 93 (6), 1155-1177. http://dx.doi.org/10.1086/261354
- [88] Short, H., Zhang, H., & Keasey, K. The link between dividend policy and institutional ownership, Journal of Corporate Finance. (2002), 8, 105–122. https://doi.org/10.1016/S0929-1199(01)00030-X
- [89] Ullah, H., Fida, A., & Khan, S. The impact of ownership structure on dividend policy evidence from emerging markets KSE-100 Index Pakistan, International Journal of Business and Social Science. (2012), 3(9), 298-307.
- [90] Tee CM, Yee ASV, Chong AL. Institutional investors' monitoring and stock price crash risk: evidence from politically connected firms, <u>Review of Pacific Basin Financial Markets and Policies</u>. (2018), 21(4), 1-35. https://doi.org/10.1142/S0219091518500285
- [91] Jiang, Fuxiu, Jiang, Zhan, Kim, Kenneth A. Capital markets, financial institutions, and corporate finance in China. Journal of Corporate Finance. (2019), 63, 1-21. https://doi.org/10.1016/j.jcorpfin.2017.12.001
- [92] Aluchna, M., Berent, T., & Kamiński, B. Dividend Payouts and Shareholder Structure: Evidence from the Warsaw Stock Exchange, Eastern European Economics. (2019), 57(3), 227-250. https://doi.org/10.1080/00128775.2019.1568196
- [93] Moin, A., Guney, Y. and El Kalak, I. The Effects of Ownership Structure, Sub-Optimal Cash Holdings and Investment Inefficiency on Dividend Policy: Evidence from Indonesia, Review of Quantitative Finance and Accounting. (2020), 44, 857-900. https://doi.org/10.1016/j.gfj.2018.06.002

- [94] Byrne, J., and T. O'Connor. Creditor rights, culture and dividend payout policy, Journal of Multinational Financial Management. (2017), 39, 60–77. https://doi:10.1016/j.mulfin.2016.12.002
- [95] Tran, Q. T., Alphonse, P., & Nguyen, X. M. Dividend policy: Shareholder rights and creditor rights under the impact of the global financial crisis, Economic Modelling. (2017), 64, 502-512 https://doi:10.1016/j.econmod.2017.03.010
- [96] Tran, M., Le, T, T., & Nguyen, X, M. "Determinants of Dividend Payout Policy in Emerging Markets: Evidence from the ASEAN Region," Asian Economic and Financial Review, Asian Economic and Social Society. (2019), 9(4), 531-546. https://doi.org/10.18488/journal.aefr.2019.94.531.546
- [97] Ajibade, A. t., & Agi, M. K. Firm characteristics and dividend policy of quoted manufacturing firms in sub-sahara Africa, Asian Journal of Economics, Business and Accounting. (2020), 13(4), 1–12. https://doi.org/10.9734/ajeba/2020/v14i330196
- [98] Akhalumeh, p., & ogunkuade, Z. Ownership structure, financial leverage and dividend policies: empirical evidence from Nigerian listed firms, Journal of Educational Realities. (2021), 11(1), 1–12.
- [99] Chindengwike, J. Nexus between financial leverage and dividend payout from manufacturing firms listed at Dar es Salaam stock exchange, Tanzania, Cogent Business & Management. (2024), 11(1). https://doi.org/10.1080/23311975.2023.2292345
- [100] Long, M. S., Malitz, I. B., & Sefcik, S. E. An empirical examination of dividend policy following debt issues, Journal of Financial and Quantitative Analysis. (1994), 29(1), 131–144.https://doi.org/10.2307/2331195
- [101] John, K., & Nachman, D. C. Risky debt, investment incentives and reputation in a sequential equilibrium, The Journal of Finance. (1985), 40(3), 863–878. https://doi.org/10.2307/2327812
- [102] Akerlof, G.A.The market for "lemons": quality uncertainty and the market mechanism, The Quarterly Journal of Economics. (1970), 84 (3), 488–500. https://doi.org/10.2307/1879431
- [103] Feizal, D, A., Sudjono & Saluy, A, B. The Effect of Profitability, Leverage and Liquidity on Dividend Policy for Construction Issuers in 2014-2019, Dinasti International Journal of Economics, Finance & Accounting. (2021), 2(2), 171-184. https://doi.org/10.38035/dijefa.v2i2

[104] Gumelar, A., Setijaningsih, H. T., & Hendrian, H. The influence of liquidity, leverage, and profitability on dividend policy in manufacturing companies listed on the Indonesian stock exchange moderated by good corporate governance, Enrichment: Journal of Management. (2024), 14(1), 48-58. https://doi.org/10.35335/enrichment.v14i1.1870

Biography



Ahmed Mohamed Sakr is a professor of Finance at Arab Academy for Science and Technology and Maritime Transport. He completed his PhD in Finance from Southern Illinois University at Carbondale in 2000.and his MBA in Finance from Central Michigan University in 2000. His Professional experience as a visiting Scholar at Eastern Illinois University from 2015 till 2016. He has participated in multiple international research collaboration projects in recent years such as: Conventional Determinants of Corporate Payout Policies in the Egyptian Stock Market in 2021.



Heba Mohamed Abdel Gawad is an assistant professor at the Arab Academy for Science and Technology and Maritime Transport, Department of Finance and Accounting. She completed her PhD in Finance in 2022 and finished her Masters of science degree in Finance in 2012 from the Arab Academy for Science and Technology and Maritime Transport. She is an academic Advisor and Course Coordinator in Finance and Accounting department since 2012 till Present.



Samar Mohamed Naguib ElBanna is a Master of science student at the Arab Academy for Science and Technology and Maritime Transport, College of Management and Technology, Finance Department. She is an Accounting and Business teacher, IGCSE Department since 2008 till present. She graduated from Faculty of commerce, Accounting department in 2004.

Research Field

Ahmed Mohamed Sakr: Corporate Finance, Accounting, Investments and Portfolio Management, International Finance, Banking.

Heba Mohamed Abdel Gawad: Corporate Finance, Accounting, Investments and Portfolio Management, International Finance, Banking.

Samar Mohamed Naguib ElBanna: Accounting and Finance.